Author's response to reviews

Title: Kaplan anastomosis of the ulnar nerve. A case report

Authors:

Georgios GP Paraskevas (g_paraskevas@yahoo.gr)
Christos ChG Gekas (gekaschris@yahoo.gr)
Alexandros AT Tzaveas (tzaveas@hotmail.com)
Ioannis IS Spyridakis (jaspyr@otenet.gr)
Alexandra AS Stoltidou (alexmed501@yahoo.com)
Parmenion PP Tsitsopoulos (par_tsits@yahoo.gr)

Version: 3 Date: 18 December 2007

Author's response to reviews:

Dear Sir or Madam,

I send you this cover letter with a point-by-point description of the changes made, regarding the submitted paper "Kaplan anastomosis of the ulnar nerve. A case report".

The changes are:

1. Following your proposal, we have excluded one scientist from the authors list (Orestis Ioannidis).
2. The manuscript has been checked over by a native English speaker and appropriate corrections were made.
3. We improved the photo quality with technical means.
4. The schematic representation of the photo has been improved.
5. We focused on our finding's clinical application in the text of "Conclusions".

I am looking forward for you reply.

Sincerely yours,

G. Paraskevas
Lecturer of Anatomy